Author's response to reviews

Title: Diagnostic accuracy of the STRATIFY clinical prediction rule for falls: A systematic review and meta-analysis

Authors:

    Jennifer Billington (jenniferbillington@rcsi.ie)
    Rose Galvin (rosegalvin@rcsi.ie)
    Tom Fahey (tomfahey@rcsi.ie)

Version: 2 Date: 11 January 2012

Author's response to reviews: see over
Dear Editorial Team,

We should be most grateful if you would consider our paper: ‘Diagnostic accuracy of the STRATIFY clinical prediction rule for falls: A systematic review and meta-analysis’ for publication in BMC Family Practice. This systematic review with meta-analysis has examined the diagnostic accuracy of the STRATIFY rule, which is commonly used in clinical practice to predict risk of falls.

We feel that this original piece of work is suitable for this journal as the issue of falls is quite topical at the moment in the medical literature. If this paper is considered for publication, we would be delighted to edit it as you see fit. All authors have read the final version of the manuscript and are happy with the content. We declare no conflict of interest in relation to the paper. We include the entire paper for your records and look forward to hearing from you.

Kind Regards,

_________________________

Dr. Rose Galvin